Author’s response to reviews

Title: An evaluation of an intervention designed to help inactive adults become more active with a peer mentoring component: a protocol for a cluster randomised feasibility trial of the Move for Life Programme.

Authors:
Andrew O'Regan (andrew.oregan@ul.ie)
Liam Glynn (Liam.Glynn@ul.ie)
Enrique Bengoechea (Enrique.Garcia@ul.ie)
Monica Casey (monica.casey@ul.ie)
Amanda Clifford (amanda.clifford@ul.ie)
Alan Donnelly (alan.donnelly@ul.ie)
Andrew Murphy (andrew.murphy@nuigalway.ie)
Stephen Gallagher (stephen.gallagher@ul.ie)
Paddy Gillespie (paddy.gillespie@nuigalway.ie)
John Newell (john.newell@nuigalway.ie)
Mary Harkin (info@ageandopportunity.ie)
Phelim Macken (pmacken@limericksports.ie)
John Sweeney (John@Claresports.ie)
Mo Foley-Walsh (Mo.foleywalsh@limerick.ie)
Geraldine Quinn (geraldinemary.quinn@hse.ie)
Kwok Ng (Kwok.Ng@ul.ie)
Nollaig O'Sullivan (nollaig.osullivan@ul.ie)
Gearoid Balfry (Gearoid.Balfry@ul.ie)
Catherine Woods (catherine.woods@ul.ie)
Author’s response to reviews:

20/06/2019

Dear Lehana,

Thank you for your quick response to our resubmission. We are grateful that you pointed out that the term ‘protocol’ should be in the title. We have added the words “a protocol for” to the title. Also, we have held onto the term ‘feasibility study’ to keep the nomenclature in the title consistent with the abstract and main text. The point by point response is below. We are hopeful that the study will be accepted for publication and thank you and your review team for the helpful feedback throughout.

Yours sincerely/ le gach dea ghuí,

Andrew

Reviewer reports:

Editor's comments:

1) Please indicate in the title that this is a protocol for a pilot trial.

Response of the authors: the title should have included the word protocol and the authors thank the editor for suggesting this. We have maintained the term ‘feasibility study’ to keep the nomenclature consistent with the abstract and main text. The new title reads:

“An evaluation of an intervention designed to help inactive adults become more active with a peer mentoring component: a protocol for a cluster randomised feasibility trial of the Move for Life Programme.”